Author's response to reviews

Title: Mothers' AdvocateS In the Community (MOSAIC) - non-professional mentor support to reduce intimate partner violence and depression in mothers: a cluster randomised trial in primary care

Authors:

Angela J Taft (a.taft@latrobe.edu.au)
Rhonda Small (r.small@latrobe.edu.au)
Kelsey L Hegarty (k.hegarty@unimelb.edu.au)
Lyndsey F Watson (l.watson@latrobe.edu.au)
Lisa Gold (lisa.gold@deakin.edu.au)
Judith A Lumley (j.lumley@latrobe.edu.au)

Version: 4 Date: 17 March 2011

Author's response to reviews:

The editors
BMC Public Health
17 March 2011

Re MS: 5470133514468596, Formatting changes required:

Major revisions

(1) Author list: have now been written exactly as they are in the submission system
(2) Abstract: is presented in the manuscript and matches exactly what is in the submission system
(3) Authors' contributions: now indicate that all competent authors have read and approved the final manuscript. E Prof Lumley has retired and is not well enough to undertake a final approval.
(4) Table 3: has been reformatted. It has been very difficult to do this, so I hope it is now OK.

Minor revisions

1. Title page: the author list, affiliations and email addresses have now been presented separately
2. Highlighting/tracking: has been removed
3. Introduction: has been renamed 'Background'.
4. Authors' contributions: have been formatted into a paragraph
5. Acknowledgements: is now a heading after the conflicting interests’ statement.
6. Blank pages: have been removed.
7. Typography: has been checked thanks

Sincerely

Angela Taft